Author’s response to reviews

Title: Involving seldom-heard groups in a PPI process to inform the design of a proposed trial on the use of probiotics to prevent preterm birth: a case study

Authors:

Juliet Rayment (juliet.rayment.l@city.ac.uk)
Christine McCourt (christine.mccourt.l@city.ac.uk)
Rosemary Lanlehin (R.M.B.I.Lanlehin@city.ac.uk)
Shahid Husain (s.m.husain@qmul.ac.uk)

Version: 1 Date: 19 Apr 2017

Author’s response to reviews:

We would like to thank the reviewers for their helpful and considered responses. We have compiled our response in an excel spreadsheet which you can find attached to this submission.